Author's response to reviews

Title: Clinical, Molecular and Drug sensitivity Pattern of Mycobacterial Isolates from extra-pulmonary tuberculosis cases in Addis Ababa, Ethiopia

Authors:

Workneh K Hirgo Mr. (workneh.korma@aau.edu.et)
Adane M Bekele Asso. Professor (adane_mihret@yahoo.com)
Jemal H Wondmagegn Dr. (jemaldr@gmail.com)
Richard M Anthony Dr (r.anthony@kit.nl)
Mekuria L Woldu Asso.Professor (yibza2002@yahoo.com)
Abraham A Armdie MD, PhD (aseffaa@gmail.com)

Version: 5 Date: 4 October 2015

Author's response to reviews: see over
Dated on 10/04/2015

Reply to the editors 3

1) Move Tables to the end of the document and remove the Figures from the main text.

2) Remove Figure titles from the uploaded Figures. The Figures and Tables could do with some editing to improve their presentation.

3) Please include the role of Workneh K Hirgo in the Author contribution section.

4) Remove author titles in the submission system.

Thank you for your comments!

Here with, I made the adjustments and corrections as per the comments

1. All the tables were listed at the end of the documents with their description. As well as the figures also submitted separately and removed from the main text.

2. The figures uploaded separately as already done before,

3. The role for workneh korma, as PI is also mentioned in the authors’ contribution section!

4. The Authors title also removed from the submission system

Regards,

Workneh Korma